Author's response to reviews

Title: Integration in primary community care networks (PCCNs): examination of governance, clinical, marketing, financial, and information infrastructures in a national demonstration project in Taiwan

Authors:

Blossom Yen-Ju Lin (yenju1115@hotmail.com)

Version: 6 Date: 16 June 2007

Author's response to reviews: see over
Dear BioMed Central Editorial Production Team:

All revisions have been made and shown point-by-point descriptions in the following pages.

Thank you so much for your help!

Sincerely,

Blossom Yen-Ju Lin
1. Please remove your job title from the affiliation section of the title page, or the affiliation will display incorrectly once the article is published.

   Ans. The job title has been removed. (p.1)

2. Please change the heading Discussions → implication for health policy makers and providers --> Discussion

   Ans. It has been changed to Discussion. (p.16)

3. Please convert the rest of the heading into a sub-heading (of appropriate font size) if you wish the rest of the heading to remain within the text.

   Ans. All the sub-headings has been converted (p.6, 9, 13, 14, 15, 16).

4. Please change Conclusion --> Conclusions

   Ans. It has been changed to Conclusions. (p.18)

5. Please insert the heading Figure legends above your figure legends, and the heading Tables above your tables.

   Ans. The heading Figure legends have been placed above the figure legends (p.27), and the heading Tables has been added above the tables (p.28).